### NBER WORKING PAPER SERIES

### VALUATION, ADVERSE SELECTION, AND MARKET COLLAPSES

Michael J. Fishman Jonathan A. Parker

Working Paper 18358 http://www.nber.org/papers/w18358

NATIONAL BUREAU OF ECONOMIC RESEARCH 1050 Massachusetts Avenue Cambridge, MA 02138 September 2012

A draft of this paper was presented in 2009 and the first half of 2010 under the name "Valuation and the Volatility of Investment." For helpful comments, we thank Andrea Eisfeldt, V.V. Chari, Gary Gorton, Igal Hendel, Alessandro Lizzeri, Andrew Nowobilski, Alessandro Pavan, George Pennacchi, Andrew Winton and seminar participants at Brown, Chicago, Northwestern, Stanford, Texas, Yale, the 2010 Minnesota Corporate Finance Conference, the 2010 SED Meetings, and the 2011 North American Winter Meetings of the Econometric Society, although we remain responsible for any errors, omissions, or typos. Luis Bryce provided excellent research assistance. The views expressed herein are those of the authors and do not necessarily reflect the views of the National Bureau of Economic Research.

NBER working papers are circulated for discussion and comment purposes. They have not been peerreviewed or been subject to the review by the NBER Board of Directors that accompanies official NBER publications.

© 2012 by Michael J. Fishman and Jonathan A. Parker. All rights reserved. Short sections of text, not to exceed two paragraphs, may be quoted without explicit permission provided that full credit, including © notice, is given to the source.

Valuation, Adverse Selection, and Market Collapses Michael J. Fishman and Jonathan A. Parker NBER Working Paper No. 18358 September 2012 JEL No. E44,G01,G2,G28

### ABSTRACT

Valuation has an externality: it creates information on which adverse selection can occur. We study a market in which investors (or lenders) buy uncertain future cash flows that are ex ante identical but ex post heterogeneous across assets from sellers (or borrowers) with reservation values. There exists a limited amount of a costly technology that can be purchased before the market opens that allows an investor to value an asset — to get a private signal of the future payoff of that asset. Because sellers of assets that are valued and are rejected can sell to other investors, there are strategic complementarities in the choice of the capacity to do valuation, the private benefits to valuation exceed its social benefits, the market can exhibit multiple equilibria, and the market can deliver a unique valuation equilibrium when it is more efficient to transact without valuation. In the region of multiplicity, the move from a pooling equilibrium to a valuation equilibrium is always socially inefficient and has many features of a financial crisis: interest rate spreads rise, trade declines, unsophisticated investors leave the market, and sophisticated investors make profits. The efficient equilibrium in the region of multiplicity can be ensured by a large investor with the ability to commit to a price. We characterize several policies that can improve on market outcomes.

Michael J. Fishman Kellogg School of Management Northwestern University Jacobs Center, Room 417 2001 Sheridan Road Evanston, IL 60208-2006 m-fishman@kellogg.northwestern.edu

Jonathan A. Parker Finance Department Kellogg School of Management Northwestern University 2001 Sheridan Road Evanston, IL 60208-2001 and NBER Jonathan-Parker@Kellogg.Northwestern.edu

# 1 Introduction

Most real investment – buying a house, starting or expanding a firm, or maintaining a business in bad times – relies on external financing, a transfers of resources today for a claim on uncertain resources in the future. Markets for external financing exhibit crises or panics in which both the volume of funding and prices collapse. Notable recent examples in the U.S. include collapses in markets for mortgages lending, for the commercial paper of financial institutions, for venture capital, and for trading and originating asset-backed securities.

In this paper, we present a theoretical model of a market for external financing and study booms and crashes based on different equilibrium levels of valuation by market participants. The model has four key ingredients. First, sellers seeking funds sell claims to future payments that are expost heterogeneous but are ex ante identical. Second, a subset of financial investors (buyers) can commit to perform a limited number of costly evaluations of these ex ante identical assets and receive signals about future payoffs. Third, sellers have a fixed reservation value for selling due for example to the possibility of not undertaking the real activity being funded. Finally, assets that are evaluated by one investor and not bought can be sold to another investor.

In this setting, we show that valuation creates asymmetric information on which adverse selection can occur, and that this externality generates strategic complementarities in valuation that can lead to multiple equilibria. The move from an equilibrium without valuation to an equilibrium with valuation has many features of a credit crunch: when valuation increases, interest rate spreads rise, lending/trade in assets declines, unsophisticated investors leave the market, and sophisticated investors make profits. Because the private benefits to valuation exceed its social benefits, the equilibria without valuation are more efficient, so that in the region of multiplicity switching from an equilibrium without valuation to one with valuation is undesirable. In terms of policy, to ensure the efficient outcome requires not just a large, unsophisticated investor, but one with the ability to commit to a price ex ante, and, in some regions of the parameter space, also a subsidy for purchasing.

More specifically, we consider a rational expectations model of a competitive market in which risk neutral sellers (real investors) sell assets at prices above their reservation values, and risk neutral financial investors compete to purchase these assets given a fixed opportunity cost of capital. Assets are ex ante identical but ex post payoffs are heterogeneous across assets. Untraded assets disappear at the end of each period so that periods are physically unconnected.

There are two types of investors. Unsophisticated investors are competitive price-takers who buy assets at their expected present discounted values. Sophisticated investors invest ex ante in capacity to perform valuation and, through choice of funds, can commit to valuing before buying any assets. Valuation capacity is costly and limited in aggregate. The use of a unit of valuation capacity provides a signal of the quality of an asset. Conditional on a good signal, an asset is worth more than the reservation value in expectation; conditional on a bad signal, it is not. Valuation is unobservable and nonverifiable, and all sellers are anonymous in the sense that, the never-valued asset is indistinguishable from the previously-valued asset to every seller except the one that performed the valuation. Thus a sophisticated investor who values an asset, observes a bad signal of the future payoff, and does not buy it, decreases the average quality of the pool of assets for other investors, which lowers the equilibrium price paid by unsophisticated investors and raises the profitability of valuation. This externality makes valuation a strategic complement.<sup>1</sup>

There is a range of parameters over which the market has multiple equilibria. In a *pooling* equilibrium no asset is valued, all assets are traded/funded, and because investors with unlimited capital compete to purchase assets, prices are high. In a valuation equilibrium sophisticated investors invest in valuation capacity, value as many assets as they can, and only good assets are sold/funded. In this valuation equilibrium, because sellers compete for limited valuation capacity, prices are low. The multiplicity is due to the strategic complementarity. The more assets are valued, the lower the average quality of unvalued assets. If the average quality falls below the seller's reservation value, unsophisticated investors leave the market, and only assets that are valued and found to be good are traded.

A switch from a pooling equilibrium to a pure valuation equilibrium matches many of the features observed in credit crunches or asset market panics.<sup>2</sup> In the pooling equilibrium (credit boom), market liquidity is high and all assets are traded/funded. In the valuation equilibrium, there is a decline in trade and observed prices fall (interest rates rise).<sup>3</sup> The price decline occurs because unsophisticated investors leave the market and market power changes from one in which assets are in short supply to one in which the ability to do valuation is in short supply. Sophisticated investors earn profits.<sup>4</sup> There is a credit crunch: in the collapse, sellers with assets (some good) that would have been sold in the pooling equilibrium are unable to get evaluated and so unable to sell in the valuation equilibrium. There is a flight to quality in two senses: only good assets are traded/funded, and unsophisticated investors leave the market and hold their funds elsewhere. Finally, because of the multiple equilibria, this shift need not be tied directly to changes in fundamentals, although changes in fundamentals can bring about the possibility of collapse and/or make collapse ultimately inevitable.

In this region of multiplicity, the benefit of valuation is not funding bad projects, yet the socially efficient outcome is the pooling equilibrium with no valuation. This follows from the fact that in the pooling equilibrium, the planner problem of whether to switch to the valuation equilibrium is (almost) the same as the sophisticated investor's problem in the decetralized market of whether to invest in the first unit of valuation capacity. Both decisions trade off funding conditional on valuation against funding without valuation given the population share of good assets. More strikingly, even in some regions where the market delivers only the valuation equilibrium, the pooling equilibrium is more efficient. This follows from the fact that in the pooling equilibrium, the presence of unsophisticated investors subsidizes investment in valuation capacity. If a seller had an asset that was valued and found to be bad and it could not go to an unsophisticated investor, it would be more averse to going to a sophisticated investor doing

<sup>&</sup>lt;sup>1</sup>In the language of Hellwig and Veldkamp (2009), the choice of the technology to uncover information is a strategic complement, but the actual information gathered is a strategic substitute, in that sophisticated investors want different information from each other (information about different assets).

<sup>&</sup>lt;sup>2</sup>Over a different region of parameters, the market can either be in a pooling equilibrium or a constrained mixed equilibrium in which sophisticated investors value as many assets as they can and buy/fund the good ones and unopshisticated investors buy/fund both the rejected assets and the un-valued assets.

<sup>&</sup>lt;sup>3</sup>In a bank run agents protect themseleves by withdrawing funds because they expect others to do so and funds are not lent to a common entity; in our model, sophisticated agents protect themseleves by investing in valuation capacity because they expect others to do so and funds are not lent to many distinct entities.

<sup>&</sup>lt;sup>4</sup>While we do not model what happens to assets once sold, in mapping to the real world, all investors potentially have mark-to-market losses on asset holdings due to the price decline.

valuation. Thus, financial market crashes of the kind just described are inefficient, even in some cases where they are driven by fundamentals and are inevitable.

Can policy correct market outcomes when inefficient? First, subsidizing trade or lowering interest rates is counterproductive, in that it actually increases the region in which the valuation equilibria is the only equilibrium and the region in which it is a possible equilibrium. Second, subsidizing the payout of bad assets reduces the region in which valuation in equilibrium is possible by reducing the economic return to separating the good from the bad. Third, a tax on valuation capacity can ensure the pooling equilibrium wherever it is efficient.<sup>5</sup>

More interestingly, because the pooling equilibrium is more efficient, a large unsophisticated investor with the ability to commit to purchase assets at the pooling price can ensure that the market selects the pooling equilibrium wherever there are multiple equilibria. Further, if the government subsidizes this large investor, it can ensure that the economy is in a pooling equilibrium wherever a pooling equilibrium would be more efficient. That said, this policy can be quite detrimental if the model if misparameterized or misapplied so that the large investor does not deter valuation and purchases at a high price previously-valued assets.

Finally, one might consider policy changes to the model environment to make valuation observable. This could eliminate the valuation externality. However, the incentive of the paired sophisticated investor and seller with a bad asset is to hide both the fact of and the outcome of valuation.

This model omits many factors that amplify and cause financial crises.<sup>6</sup> In what circumstances is its logic more likely to apply? Our model captures the market for assets that are ex ante equivalent and the financial fluctuations in the model are increases or decreases in valuation, and so increases or decreases in market segmentation and market depth. While some valuation occurs in all markets for external finance, every asset is valued up to some point, and then pooled with observationally equivalent assets. For our model to explain a part of a financial crises, it must be that one market is particularly important – as for example the market for AAA commercial paper or the mortgage market – or equilibrium selection must be correlated across markets. And the model is most relevant when bad signals are rare, as in markets for issuance either of highly-rated debt that is ex ante unlikely to default or of equity for new, firms that are almost sure to make productive use of new funds (e.g. the dot com or IPO markets).

We see two situations where the model may prove useful. First, in new markets there is little record on the performance of new types of assets and the investment associated with them. The cost of valuation for some information is initially high and the market is necessarily in a pooling equilibrium with respect to such information. As different assets/investments are observed to profit or fail, valuation costs may decline over time, and as valuation costs decline, the collapse to the valuation equilibrium becomes possible and ultimately inevitable. In this case, the precursors to collapse are the two main factors identified by Kindleberger (2000): credit – worsened by leverage which is not present in the model – and displacement – a new technology or investment opportunity. Second, in markets where initially all assets are good, the market

<sup>&</sup>lt;sup>5</sup>It is important to note that the model omits many benefits of valuation, most notably that valuation might reveal information about the aggregate payoff of all assets.

<sup>&</sup>lt;sup>6</sup>A short list of important papers focussed on omitted factors: Diamond and Dybvig (1983) Bernanke and Gertler (1989), Kiyotaki and Moore (1997), Holmstrom and Tirole (1998), Caballero and Krishnamurthy (2003, 2008), and Geanakoplos and Fostel (2009).

is automatically in a pooling equilibrium. But since there is no incentive to produce assets of higher quality among assets that are sold at identical prices, the share of good assets may naturally decline over time, which again makes a collapse to the valuation equilibrium possible and possibly inevitable.<sup>7</sup>

The paper is organized as follows. The next section describes the related literature not discussed elsewhere and Section 3 presents the model. Section 4, using two lemmas, derives the value functions as a function of equilibrium variables. Section 5 contains a characterization of the equilibria of the model. Section 6 gives our main result on multiplicity and efficiency of equilibria, interprets an equilibrium switch as a contributor to a financial crisis, and discusses the importance of several modelling assumptions. Section 7 contains analysis of five possible policy interventions to correct market inefficiencies and a final section concludes.

# 2 Related literature

Our main contributions – the multiple equilibria, booms/busts, and the welfare and policy implications – stem from endogenous acquisition of information possibly leading to market collapse due to adverse selection and so build directly on Akerlof (1970)'s seminal model and Hirshleifer (1971)'s insights about the timing of the arrival of information. The key to our results is the type of information acquired. Most work on information acquisition and the trading and pricing of financial assets assumes that the information acquired either reveals and makes into common knowledge information that which was previously private information and so solves problems of asymmetric information, or alternatively provides information about payoffs that are common across assets and so is revealed by trade and underproduced from a social perspective. In contrast, our key modelling choice is that valuation reveals information about a payoff that is common across agents but specific to the particular asset.<sup>8</sup> This sets our paper apart from the literature on the strategic revelation of private information (e.g. Berk and Uhlig (1991)), the literature on the incentive to value assets (and the Grossman-Stiglitz paradox), and the literature using costly state verification.<sup>9</sup> Relative to previous work on adverse selection in lending markets, we focus on parameters where the random, unvalued assets (loan) is worth buying (making) at a given market price and so the market can be operative without valuation.

Our results that the capacity to do valuation is a strategic complement and that valuation has a negative externality are closely related but distinct from previous insights in Dang (2008). Dang (2008) shows that increased information acquisition on one side of a trade increases the incentive to acquire information by the agent on the other side. In this sense, both papers contribute to research on the value of private information.<sup>10</sup> And while we address

<sup>10</sup>See for example Angeletos and Pavan (2007), Mackowiak and Weiderholdt (2009), and Myatt and Wallace

 $<sup>^{7}</sup>$ Both situations, and the model, echo some of the features of the 2007-2008 financial crisis. See Gorton (2010) for example.

<sup>&</sup>lt;sup>8</sup>An interesting paper with a similar approach to information but a different structure is Broecker (1990) in which valuation is costless but noisy, and correlated across lenders. In equilibrium, the winners curse from noisy valuation interacts with the adverse selection problem to generate an equilibrium with a continuum of interest rates across different banks.

 $<sup>^{9}</sup>$ Models of costly state verification (Raviv (1975) and Townsend (1979)) typically use the verification technology to make what had been private information into common knowledge, so that valuation is used to eliminate problems associated with asymmetric information.

different topics and use different screening technologies (costly state verification vs. screening with contracts), our paper is closely related to the contemporaneous paper Bolton, Santos, and Scheinkman (2011) which considers the long-term rise of the financial sector. In Bolton, Santos, and Scheinkman (2011), agents choose to become sophisticated as a career, and then compete with unsophisticated exchanges. Unlike in our model, sellers know their type and take unobserved actions leading to a moral hazard problem.

Our paper is far from the first to study fluctuations in the strength of adverse selection in financial markets; notable examples include Mankiw (1986), Eisfeldt (2004), Kurlat (2010), and Morris and Shin (2012).<sup>11</sup> In general, our contribution is to make the information on which adverse selection occurs endogenous, and as such it builds on the important papers Ruckes (2004) and Dell'Ariccia and Marquez (2006), which both consider how adverse selection and lending standards leads to contractions of credit in bad times.<sup>12</sup> The former shows how the probability that a borrower is of a bad type changes the private value of information which in turn is amplified through the winners curse. The latter focusses not on the creation of information but rather on contract terms (specifically collateral requirements) and how these change in response to exogenous changes in the share of new projects, about which no bank has information, and existing projects, about which some bank has private information. In the model, fewer new projects implies a lower share of good projects approaching banks and a tightening of lending standards and reduced credit.<sup>13</sup>

Finally, an important subset of this literature is that related to security design and securitization that focusses on how informed issuers destroy information to create a pooling equilibrium (Gorton and Pennacchi (1990), DeMarzo (2005), and DeMarzo and Duffie (1999)). Marin and Rahi (2000) considers security design and shows how the optimality of complete vs. incomplete markets (complete vs. incomplete revelation of private information) depends on the costs of adverse selection on private information relative to the costs of reduced ex ante insurance. More closely related to our paper, Dang, Gorton, and Holmstrom (2009) considers security design when the aftermarket may have valuation that hinders trade (as in Dang (2008)). And Pagano and Volpin (2008) consider security design and then a later equilibrium in which trade occurs in a noisy rational expectations equilibrium so that information can get rents. While the model exhibits an externality from valuation, it does not generate multiplicity of equilibrium (beyond those always possible in noisy rational expectations equilibria (Breon-Drish (2011))), nor is there a role for commitment and so optimal policy is different.

# 3 The model

We consider a model in which periods are physically unconnected over time and focus on equilibria in the static game.<sup>14</sup> In this section we focus only on one period. In later sections we consider dynamics with exogenously changing parameters.

<sup>(2009).</sup> Veldkamp (2011) provides an excellent and broad discussion of this literature.

<sup>&</sup>lt;sup>11</sup>All of which in turn owe much to Stiglitz and Weiss (1981) and Rothschild and Stiglitz (1976).

<sup>&</sup>lt;sup>12</sup>More recently, Gorton and Ordonez (2012) studies fluctuations in informed lending in secondary markets.

<sup>&</sup>lt;sup>13</sup>There is also work on bank credit cycles, such as Gorton and He (2007) which studies changes in valuation driven by collapse of collusive equilibria among banks (ala Green and Porter (1984)).

<sup>&</sup>lt;sup>14</sup>In Fishman and Parker (2012), we study a model in which assets do not dissapear if unpurchased.

At the beginning of the period, a unit mass of risk-neutral sellers (real investors) each enter the market seeking to sell a risky claim to a real investment to a large number of competitive risk-neutral financial investors each with access to unlimited funds at constant gross interest rate R > 1. Each seller has one asset of fixed size, has no funds, and has a reservation price of 1. The asset if sold pays out a random amount D in the future and the seller must sell all the asset or none of it. The most natural interpretation of this setup is that the seller has a project that only it can undertake, that costs 1, and that therefore is only worth doing if it can sell the future stream of profits for more than the cost to undertake the project.<sup>15</sup> All sellers and investors initially have common expectations of the future payout of the asset, E[D].

Sellers are anonymous: within the period, a seller can visit a finite number of investors anonymously and simultaneously so that a seller turned away from one investor is able to go to another investor and appear indistinguishable from any other seller.<sup>16</sup> Sellers that sell their assets leave the market at the end of the period and sellers with unsold assets disappear at the end of the period. Thus periods are not physically connected.

There are two types of investors. Unsophisticated investors cannot do valuation and have a flexible amount of funds. Sophisticated investors must choose at the beginning of the period (before any buying or selling) both how much capital to raise to purchase assets (f for funds) and how much valuation technology to acquire (h for human capital).<sup>17</sup> The cost of a unit of valuation capacity is c up to  $\chi_i$  for sophisticated investor i, and infinite thereafter, so  $h_i \leq \chi_i$ .<sup>18</sup> We denote the aggregate amounts of funding and valuation capacity by F and H respectively, and the aggregate constraint on total valuation capacity by  $\bar{\chi} < 1$ , so

$$H \le \bar{\chi}.\tag{1}$$

One unit of valuation technology allows the valuation of one asset which reveals a binary signal of the quality of that asset. The expected payoff of an asset is  $D^g = E[D|good]$  conditional on a good signal and  $D^b = E[D|bad]$  conditional on a bad signal. A good asset is worth investing in/buying and a bad asset is not:

$$D^g > R > D^b$$

The population share of assets that are good, is  $\lambda \in (0, 1)$ , so that

$$E[D] = \lambda D^g + (1 - \lambda) D^b$$

<sup>&</sup>lt;sup>15</sup>An alternative interpretation is that the asset is an existing asset, but that the seller has limited exposure to the downside risk, for example due to bankruptcy, their particular compensation contract, or a bailout. In this alternative interpretation, if there is no social benefit to not selling bad assets, an equilibrium without valuation is more efficient for all parameter values.

<sup>&</sup>lt;sup>16</sup>This assumption is a static analog to a continuous process with valued projects indistinguishable from new entrants. A similar mechanism is used in Zhu (forthcoming) in which sellers pursue sequential search but buyers do not know in what order the seller has gone to potential buyers. And to be clear, implicit in this assumption is the inability of sellers with unvalued projects to commit to delay sale ala Guerrieri and Shimer (2012) and Chang (2011).

<sup>&</sup>lt;sup>17</sup>The choice of funding capacity is a device to allow buyers to commit to do valuation. This commitment is not infrequently simply assumed in the cited literature.

<sup>&</sup>lt;sup>18</sup>The constraint on valuation could instead be an assumed to be a limit on the amount of financial capital available to the sophisticated investors at the start of the period.

The outcome of valuation is observed by both the investor and seller, but is not observable by other investors or sellers. Investors are not anonymous: market participants can observe available funds, investment in valuation technology, and the prices of transactions.

Each period unfolds as follows. First, sophisticated investors choose valuation technology, then the market opens: i) each seller approaches a buyer/investor and either sells or repeats the process by continuing to another buyer; ii) unsophisticated act competitively to buy/fund assets; and iii) sophisticated investors use their funds and valuation technology to value and buy assets.

To complete the model we need to specify how prices are determined for a matched sophisticated investor and seller. For simplicity, we assume that when information creates a joint surplus for a matched investor-seller, the transaction price is set so that the entire surplus is captured by the investor.<sup>19</sup>

We rule out reputational equilibria and consider only the subgame perfect Nash equilibria of the period game. Sophisticated investors maximize profits by first choosing valuation capacity and funds taking as given their own future behavior, the strategies of other sophisticated investors, and the price that will be available from unsophisticated investors. Subsequently, sophisticated investors use funds and valuation technology and choose to buy and at what price taking as given the strategies of other agents and the price available from unsophisticated investors. Sellers do the same for their choices of what investors to approach and at what prices to sell. We assume that investors randomize across equivalent sellers and sellers randomize across equivalent investors and a law of large numbers allows us to ignore uncertainty from this randomization.

We discuss the importance of various assumptions not discussed elsewhere in Section 6.4. The next section derives value functions and the following section describes equilibria.

# 4 Equilibrium value functions

This section contains two subsections that characterize behavior and prices sufficient to derive the value functions for sellers and buyers: a first subsection for sophisticated investors and sellers approaching them, and a second subsection for unsophisticated investors and sellers approaching them. The description of the equilibria in the Introduction is useful for following the initial analysis of this section.

We denote the equilibrium price paid by a sophisticated investor for a good asset by  $P^g$  and for a bad asset by  $P^b$ , and the equilibrium price paid by an unsophisticated investor by  $P^U$ .

#### 4.1 Sophisticated investors

Consider first a sophisticated buyer/investor matched with an asset that it has valued. The potential buyer values the asset at  $D^i/R$  for i = b or g, and the seller has three options besides selling: i) go to another sophisticated investor; ii) go to an unsophisticated investor; iii) keep

<sup>&</sup>lt;sup>19</sup>That is, the investor has all the bargaining power in a Nash bargaining situation. An alternative assumption with the same implications is that sophisticated investors post prices at which they are then committed to transact if they transact.

the asset and get 1. The buyer buys the asset as long as the value of the asset is greater than the largest of these three outside options at a price equal to this largest outside option.

First note that if the asset was found to be bad, there is no price at which this asset is purchased since the last outside option – keeping the asset – gives greater value than the value of the asset  $(D^b/R-1 < 0)$ . Further, if the asset was found to be bad, leaving this seller destroys the value of the good information the seller and buyer share. Because valuation capacity is always insufficient to value all assets and because sellers are anonymous and valuation is nonverifiable, sophisticated investors do not compete for valued assets and do not bid prices above the better of the other two options.<sup>20</sup> Thus, provided the sophisticated investor has sufficient funds, an asset found to be good is purchased at

$$P_i^g = P^g = \max\left[P^U, 1\right] \tag{2}$$

and sellers with bad assets continue to search for buyers. The rejected seller has two remaining options: i) take its reservation value, 1; ii) go to an unsophisticated investor and sell for  $P^{U}$ . Summarizing this we have the following lemma.

**Lemma 1** A sophisticated investor matched with an asset that it has valued (i) buys/funds the asset at  $P^g = \max [P^U, 1]$  if it is good and  $f > P^g$ . (ii) does not buy/fund the asset if it is bad.

It is now useful to partially characterize  $P^U$ . Since unsophisticated investors compete to fund projects, they set prices equal to the present discounted value of dividends. When there is no valuation, this implies  $P^U = E[D]$ . When there is valuation, good projects are removed from this pool, and we claim (verified below) that  $P^U < E[D]$ .

Next, we turn to the question of the choice of funds by the sophisticated investor. If no valuation capacity is purchased, then we assume zero funds are chosen.<sup>21</sup> If valuation capacity is positive, a sophisticated investor chooses funds equal to the cost of purchasing all the good assets it would find using all its valuation capacity when only un-valued sellers approach it, which is  $P^g \lambda h_i$ .<sup>22</sup> It buys no bad assets and no assets without valuation. Why? If the sophisticated buyer chose less than this amount of funds, costly valuation capacity would go to waste. If it had funds to spare after funding good projects, it could buy some assets without valuation or buy some bad assets. Buying a random un-valued asset from the population of assets is strictly preferred whenever some valuation is happening because  $E[D] > P^U$ . But then, knowing this, sellers that have assets that they know are bad would approach this investor the same way they might approach an unsophisticated investor and the sophisticated investor would no longer be drawing a random asset from the population but instead from a pool of unvalued assets valued and those valued and found to be bad. Thus, a sophisticated investor that chooses funds sufficient to buy unvalued assets would reduce the quality of its pool of applicants and so reduce

<sup>&</sup>lt;sup>20</sup>It is straightforward to confirm that there is surplus:  $P^U < D^g/R$  or unsophisticated investors lose money, so that  $P^g < D^g/R$  for  $\lambda < 1$ .

<sup>&</sup>lt;sup>21</sup>While it is possible that a sophisticated investor that does not invest in valuation capacity mimics an unsophisticated investor and chooses a large amount of funding capacity, this does not affect the equilibrium price or quantity and we ignore this for ease of exposition.

<sup>&</sup>lt;sup>22</sup>Sophisticated investors value a positive measure of assets and find a nonstochastic share of good projects.

the efficiency of its use of valuation, and would do so sufficiently as to be not worth doing. These results are summarized in the following lemma, proved in appendix A (following the above logic).

# Lemma 2 (Funding with valuation, funding capacity, and share of good assets)

If  $h_i > 0$ , the sophisticated investor chooses  $f_i = P^g \lambda h_i$ , gets a share  $\lambda$  of good assets, only buys after valuation finds the asset to be good, and uses all its funding capacity.

From here on, we refer only to valuation capacity since funding capacity is equal to valuation capacity.<sup>23</sup>

We can now write the value of a unit of sophisticated capital. The value of investing in a unit of valuation technology and an associated unit of funding capacity is

$$J^{S} = -c + \lambda \left(\frac{D^{g}}{R} - \max\left[P^{U}, 1\right]\right).$$
(3)

This equation is linear and decreasing in the one endogenous variable, the market price. Taking  $P^U$  as given, if  $P^U$  is such that  $J^S > 0$ , then sophisticated investors purchase valuation capacity up to the constraint,  $\bar{\chi}$ , that is:  $H = \bar{\chi}$ . Similarly if  $P^U$  is such that  $J^S < 0$ , then H = 0, and if  $J^S = 0$ , then  $H \in [0, \bar{\chi}]$ . Thus, the model will have regions of multiple equilibria as long as the price  $P^U$  decreases in the aggregate use of valuation.<sup>24</sup>

Turning to sellers, the (net) expected value to the (uninformed) seller of going to a sophisticated investor is

$$W^{S} = \lambda P^{g} + (1 - \lambda) \max\left[P^{U}, 1\right] - 1 \tag{4}$$

where the max term reflects the fact that the seller found to have a bad asset chooses between keeping the asset or selling the asset to an unsophisticated investor.

It is worth pausing to note that, given our assumptions, investors would like to use contract terms to screen assets and save on valuation capacity. This could be done by sophisticated investors with an ex ante fee, which in the model is ruled out twice. First, we assume that sellers have no funds: since no sellers have any funds, an application fee would make no profits and buy no assets. Second, we assume that after valuation, investors have all the power in the bargaining relationship. Given this, after valuation, the best price a seller can hope for is the market price (or reservation value), meaning that no seller would pay a fee in equilibrium.<sup>25</sup> An alternative screening mechanism would be to allow investors to impose a penalty on the seller whose asset pays off poorly – have the seller bear some risk. Two common foundations for the assumption that the seller must sell all the asset are either moral hazard on the part of the new owner (investor) or there being no resources for the investor to collect if the asset turns out to be bad.

 $<sup>^{23}</sup>$ While it is possible that a sophisticated investor that does not invest in valuation capacity mimics an unsophisticated investor and chooses a large amount of funding capacity, this does not affect the equilibrium price or quantity and we ignore this for ease of exposition.

<sup>&</sup>lt;sup>24</sup>And will for more general cost functions as long as cost is not increasing faster than price is decreasing.

<sup>&</sup>lt;sup>25</sup>Relaxing both these assumptions and allowing a fee, investors will generally find it profitable to choose more funds and do stochastic valuation, funding some projects without valuation. We conjecture than equilibria of similar flavor exist in a model in which valuation is noisy and the fee is capped due to the inability to commit to a share of the surplus (or due to the possibility of other agents mimicing sophisticated investors and charging a fee but not funding any applicants). But we have also found models in which no equilibria exist (for similar reasons as in insurance markets).

#### 4.2 Unsophisticated investors

Since all valuation capacity is used, the aggregate share of assets that are valued in equilibrium is H. Of these,  $\lambda H$  are found to be good and so are purchased by sophisticated investors. The total number of assets remaining is the sum of the 1 - H assets that are not valued and the  $H(1 - \lambda)$  that are valued and found to be bad, so that the share of assets that are good and seek to sell without valuation is

$$\frac{\lambda \left(1-H\right)}{1-\lambda H}.$$
(5)

When no assets are valued, H = 0, and this equals the population share of good assets,  $\lambda$ .

We denote by  $J^U$  an unsophisticated investor's value of a buying an asset without valuation. This value is the expected discounted payout of the asset less the price paid for the asset

$$J^{U} = \frac{\left(\frac{\lambda(1-H)}{1-\lambda H}\right)D^{g} + \left(1 - \frac{\lambda(1-H)}{1-\lambda H}\right)D^{b}}{R} - P^{U}$$

For a seller (with or without information about its asset's value), the (net) value of going to an unsophisticated investor is

$$W^U = \max\left[P^U - 1, 0\right] \tag{6}$$

which is also the social surplus of this transaction.

Price competition among unsophisticated investors leads to zero-profits in equilibrium,  $J^U = 0$ , which implies that the price paid by the unsophisticated investors is:

$$P^{U}(H) = \frac{\lambda \left(1 - H\right) D^{g} + \left(1 - \lambda\right) D^{b}}{\left(1 - \lambda H\right) R}$$

$$\tag{7}$$

Thus, unsophisticated investors set the market price as a function of the average quality of assets they face in equilibrium. If that price is below the reservation value of sellers, then they do not purchase any assets.

To complete our specification, denote the market price at which transactions occur by

$$P = Max \left[ P^U, 1 \right] \tag{8}$$

Equation (7) and the value function of the sophisticated investors, equation (3), illustrate the main externality in the model. The profits of the sophisticated investors are decreasing in  $P^U$  which in turn is decreasing in the aggregate amount of valuation capacity purchased. More valuation worsens the pool of asset purchased by unsophisticated investors, which lowers the price they are willing to pay, which lowers the price that sophisticated investors have to pay for good assets, which makes valuation more profitable.

### 5 Equilibria

Equilibria can now be characterized using equations (2), (3), (4), (6), (7), (8), and our characterization of optimal choice of H.

There are four types of equilibria: pooling, valuation, constrained mixed, and unconstrained mixed.

#### 5.1 The pooling equilibrium

In a pooling equilibrium all assets trade without valuation at the same price and no valuation technology is used. For this equilibrium to exist, a sophisticated investor must find it unprofitable to invest in valuation capacity and uninformed sellers must prefer selling to unsophisticated investors to keeping the asset, both when H = 0:

$$J^S \leq 0$$
$$W^U \geq 0$$

The second inequality implies  $P^U - 1 \ge 0$  (when H = 0), so that with equation (7) these conditions can be written as

$$c \geq \lambda (1-\lambda) \frac{D^g - D^b}{R}$$

$$\lambda \geq \frac{R - D^b}{D^g - D^b}$$
(9)

and all assets trade, so volume is 1, at price equal to the unconditional expected value,

$$P = \frac{\lambda D^g + (1 - \lambda) D^b}{R} = E[D] / R.$$

The pooling equilibrium exists as long as i) the marginal cost of the valuation technology is large enough relative to the gain from valuation, and ii) the population expected return without valuation is high enough. Note that the right hand side of the first inequality is equal to the probability of the asset being good ( $\lambda$ ) times the joint gain in value when it is good  $((1-\lambda)\frac{D^g-D^b}{R} = \frac{D^g}{R} - \frac{\lambda D^g + (1-\lambda)D^b}{R} = \frac{D^g}{R} - P)$ , which is the private value of information at the margin in the pooling equilibrium.

#### 5.2 Equilibria with valuation

There are three possible types of equilibria in which investors invest in the valuation technology. First, there is a valuation equilibrium in which sophisticated investors value and buy as many good assets as they can and make profits, and the residual pool of assets is so poor on average that unsophisticated investors do not buy assets. Second, there is a constrained mixed equilibrium which is like the valuation equilibrium except that the residual pool of assets is good enough on average that the remaining assets are purchased by unsophisticated investors. Finally, there is an unconstrained mixed equilibrium in which sophisticated investors invest in some valuation capacity  $\langle \bar{\chi}, \rangle$  perform valuation and invest in some assets, unsophisticated investors buy the remaining assets, no investors make profits, and all uninformed sellers are indifferent between investors.

#### 5.2.1 The valuation equilibrium

For the valuation equilibrium to exist, each sophisticated investors must prefer to invest in valuation up to its capacity constraint, and each uninformed sellers must prefer to go to a

sophisticated investor or keep its asset instead of going to an unsophisticated investor, both when  $H = \bar{\chi}$ :

$$\begin{array}{rccc} J^S & \geq & 0 \\ 0 & > & W^U \end{array}$$

Since  $0 > W^U$  implies  $P^U < 1$ , we have that  $P^S = 1$ , and these conditions become

$$c \leq \lambda \left(\frac{D^g}{R} - 1\right)$$

$$\lambda \leq \frac{R - D^b}{(D^g - D^b) - \bar{\chi} (D^g - R)}$$
(10)

and only good assets that have been valued trade, so volume is  $\bar{\chi}\lambda < 1$ , at price equal to the reservation value, P = 1.

The valuation equilibrium exists as long as i) the marginal cost of valuation is low enough relative to the gain from valuation, which is the probability that transaction occurs times the gain from transacting rather than the seller keeping the asset and ii) the share of good assets is low enough (or  $\bar{\chi}$  high enough) that buying with,out valuation is not profitable after  $\bar{\chi}\lambda$  good assets are bought by sophisticated investors.

It is worth noting that in the pooling equilibrium (equation (9)), the benefit of a marginal unit of valuation is the ability to avoid the unvalued assets that are bad with probability  $1-\lambda$ . In the valuation equilibrium, there is an additional benefit, the ability to avoid all assets that have previously been found to be bad by others. Thus, as the share of good assets in the population  $(\lambda)$  increases to 1 (and  $\bar{\chi}$  near one), the valuation equilibrium can occur for higher valuation costs (the first equation (10)) even though in aggregate the information gained by valuation in equilibrium is vanishing. This previews one of the results of section 6.2, that valuation can be socially inefficient but privately optimal.<sup>26</sup>

#### 5.2.2 The constrained mixed equilibrium

In the second possible equilibrium with valuation, sophisticated and unsophisticated investors both purchase assets. While as in the valuation equilibrium, sophisticated investors are at capacity and make profits, here valuation capacity is so limited or the share of good assets so high that the remaining, unvalued assets still have positive expected net present value and are bought without valuation by unsophisticated investors. As above, sophisticated investors have market power and earn the rents of valuation, but now compete with unsophisticated investors rather than the seller's outside option. In this equilibrium, profit maximization implies that uninformed sellers are indifferent between types of investors. Thus, for  $H = \bar{\chi}$ 

$$J^S > 0 \tag{11}$$
$$W^S = W^U \ge 0$$

<sup>&</sup>lt;sup>26</sup>There is a discontinuity (outside our assumed range) at  $\lambda = 1$ , where the valuation equilibrium cannot occur for c > 0 even when  $\chi = 1$ .

The inequality  $W^U \ge 0$  implies  $P^U \ge 1$  which places an upper bound on H

$$H \le \tilde{H} := 1 - \frac{(1-\lambda)\left(R - D^b\right)}{\lambda\left(D^g - R\right)} \tag{12}$$

which implies that this equilibrium requires  $\bar{\chi} \leq \tilde{H}$ . With  $P^U \geq 1$ , the equality  $W^S = W^U$  implies

$$P^{g} = P^{U} = P = \frac{\lambda (1 - \bar{\chi}) D^{g} + (1 - \lambda) D^{b}}{(1 - \lambda \bar{\chi}) R}.$$

Given this price,  $J^S > 0$  and  $\bar{\chi} \leq \tilde{H}$  simplify to the two conditions for the equilibrium to exist:

$$c < \frac{1}{1 - \lambda \bar{\chi}} \lambda (1 - \lambda) \frac{D^g - D^b}{R}$$

$$\lambda \geq \frac{R - D^b}{(D^g - D^b) - \bar{\chi} (D^g - R)}$$
(13)

The first inequality is the reverse of a 'scaled up' (by  $\frac{1}{1-\lambda\bar{\chi}}$ ) version of the first inequality for the pooling equilibrium. That is, costs have to be low enough so that valuation is profitable, and the 'scaling' factor represents the difference between the temptation to purchase valuation when no other agent does (and prices are high) and the purchase of the last unit when all other agents purchase valuation (and so prices are lower). The second inequality states that the share of good assets is high enough (or  $\bar{\chi}$  low enough) that transacting without valuation is profitable after  $\bar{\chi}\lambda$  good assets are bought by sophisticated investors. As  $\bar{\chi} \to 1$ , this lower bound on  $\lambda$ goes to 1. It is the exact complement to the second equation for the pure valuation equilibrium.

#### 5.2.3 The unconstrained mixed equilibrium

In the final type of equilibrium, for some  $H \in (0, \bar{\chi}]$ , all uninformed sellers are indifferent between sophisticated and unsophisticated investors and sophisticated investors are indifferent between investing in more capacity and not. In appendix B, we show that this unconstrained mixed equilibrium exists for parameters such that the pooling equilibrium exists and either the valuation equilibrium or the constrained mixed equilibrium exists.

We do not focus on this equilibrium because it is 'unstable' in the sense that if a sophisticated investor invested in more valuation capacity, it would reduce the quality of the assets bought by the unsophisticated investors, valuation would make more profits, and all sophisticated investors would like to have invested in more capacity to do valuation. Similarly, a slightly higher share of assets choosing to use unsophisticated investors would raise the unsophisticated market price, raising  $P^S$ , and all sophisticated investors would prefer not to have invested in capacity to do valuation.

### 6 Analysis

We first formally state our main results that there are regions of multiple equilibria, then rank them by efficiency, and finally, turn to the dynamics of a crash from the pooling equilibrium to a pure valuation or to a constrained mixed equilibrium.

#### 6.1 Regions of multiple equilibria

The analysis of the previous section implies the following theorem.

### Proposition 1 (Multiple equilibria) In any period,

(i) the region of parameters in which the valuation equilibrium can exist overlaps the region of parameters in which the pooling equilibrium can exist

(ii) The region of parameters in which the constrained mixed equilibrium can exist overlaps the region in which the pooling equilibrium can exist

(iii) The union of these two regions of multiplicity define the set of parameters in which the unconstrained mixed equilibrium exists.

**Proof.** There are allowable parameters that satisfy equation (9) and equation (10) and allowable parameters that satisfy equation (9) and equation (13). Part (iii) is proved in appendix B.  $\blacksquare$ 



Figure 1: The regions for the pooling, valuation, and constrained valuation equilibria

Figure 1 plots the areas in which each equilibrium exists in  $\lambda - c$  space (and for R = 1.1,  $D^g = 1.14$ ,  $D^b = 1.09$ , and  $\chi = 0.90$ ). When the cost of valuation is low enough, only equilibria with valuation exist. When it is high enough, only the pooling equilibrium is possible. When the share of good assets is low enough, no equilibria or only the pure valuation equilibrium exists. When the share of good assets is large enough, only the pooling equilibrium exists. For intermediate costs of valuation and an intermediate share of good assets, multiple equilibria exist.

### 6.2 Efficiency of equilibria

We define efficiency as maximizing social surplus: the sum of the value functions of the unit mass of sellers and all investors who buy assets. In this subsection, we first show that in the regions of multiplicity, the socially efficient outcome is always the pooling equilibrium. More strikingly, even in some regions where the market delivers only an equilibrium with valuation, it would be more efficient to buy/fund all assets without valuation (Pareto superior with transfers). This second region occurs because the market has a tendency to produce too much information due to the externality that valuation decreases the average quality of assets purchased by unsophisticated investors.

Consider first the parameter set for which the unconstrained mixed equilibrium exists. The pooling equilibrium exists wherever this equilibrium exists. And the mixed equilibrium is clearly inefficient since all assets are traded, as in the pooling equilibrium, but in addition some valuations are done at cost c per unit of valuation capacity.

Second, consider the parameters set for which the constrained mixed equilibrium exists. Since again the pooling equilibrium leads to all assets trading but without the costs of valuation, the pooling equilibrium is more efficient than the constrained mixed equilibrium.

Third, consider the parameter set for which the pure valuation equilibrium exists and suppose that agents were playing their pooling equilibrium strategies, so no investors purchase the capacity to do valuation. We show that the parameter set over which a sophisticated investor would choose to invest in valuation capacity is strictly greater than the parameter set over which a social planner would invest in valuation technology if she could ensure that assets known to be bad were not funded/sold.<sup>27</sup> Thus, where the market can deliver either equilibrium, the pooling equilibrium is more efficient. And in a subset of the parameter space where the market delivers only the valuation equilibrium, the planner would like to prohibit valuation.

To develop intuition, first assume that  $\bar{\chi}$  is arbitrarily close to one, so there is no inefficiency in the valuation equilibrium from not being able to value all assets. The planner would like to invest in a unit of valuation capacity only if the cost, c, is less than the expected social benefit. This benefit is the probability in the population that any given asset is bad  $(1 - \lambda)$ , times the gain from not trading it, which in turn is the reservation value of the seller less the present value of the asset  $(1 - D^b/R)$ .<sup>28</sup> Given linearity (and  $\bar{\chi}$  almost one), if the planner would chooses to value one asset, it would choose to value all assets and so the valuation equilibrium would be more efficient.

Now consider a sophisticated investor choosing whether to invest in a unit of valuation or instead to mimic an unsophisticated investor and buy/fund one asset without valuation. In either cases, if the asset is good, the investors buys it at the market price P. The private cost of valuation capacity is the same as in the social planner's problem, c. The expected private benefit is the population probability that an asset is bad – again, as in the social planner's problem – times the gain to this sophisticated investor of not buying it, which is the market price less the payout of the bad asset  $(P - D^b/R)$ , which is greater than the benefit in the planner's problem since the pooling price is greater than or equal to the reservation value  $(P \ge 1)$  for

<sup>&</sup>lt;sup>27</sup>Such as if anonymity could be destroyed by announcing that a valuation was done. As discussed in Section 7, such an announcement technology would not be used by private agents since it destroys match value.

<sup>&</sup>lt;sup>28</sup>There is no gain associated with good projects since they are funded in both equilibria.

any parameters in which the pooling equilibrium exists. Thus, the planner prefers the pooling equilibrium for all parameter values for which a sophisticated investor acting alone does not undermine it – that is for all parameter values where it exists, including the region of multiplicity.

Further, for some parameter values for which only the pure valuation equilibrium exists, trading all assets without valuation is more efficient. Why? Because valuation allows investors to avoid buying/trading bad assets. The cost of buying/trading a bad asset is the effective price paid less the payout, where the effective price is the market price for the sophisticated investor but only the reservation price for the planner. Thus, the existence of transactions without valuation at P > 1 makes valuation worth more to sophisticated investors than to the planner, which, for some parameters, undermines the existence of the pooling equilibrium where it would otherwise be more efficient.

When  $\bar{\chi} < 1$ , the argument must account for the additional inefficiency of the pure valuation equilibrium that some unvalued assets that have positive expected surplus are not traded/funded. In the pure valuation equilibrium, since sellers are all at their reservation values, total social surplus is given by the sum of the profits of the valuation done by sophisticated investors:

$$\bar{\chi}\left(-c+\lambda\left(\frac{D^g}{R}-1\right)\right)$$

If instead all assets are traded without valuation, investors all make zero profits and total social surplus is given by the total payouts to the unit mass of sellers:

$$\frac{\lambda D^g + (1-\lambda) D^b}{R} - 1.$$

Subtracting gives that no-valuation and having all assets sold/funded is socially preferred to the valuation equilibrium whenever the total cost of valuation capacity and the cost of not trading the unvalued assets exceeds the benefits of not buying the valued assets found to be bad:

$$\bar{\chi}c + (1 - \bar{\chi})\left(\frac{\lambda D^g + (1 - \lambda)D^b}{R} - 1\right) \ge (1 - \lambda)\bar{\chi}\left(1 - \frac{D^b}{R}\right)$$

As shown in Figure 2, the lower bound of this region is the line (in  $\lambda - c$  space) that runs from the point on the boundary between the pure valuation and pooling equilibria where P = 1to the maximum  $\lambda$  where the pure valuation equilibrium exists and c = 0 (where P = 1 also).

To sum up, we state these results formally.

#### **Proposition 2** Ranking of equilibria

i) For parameters such that the pooling equilibrium exists, it is more efficient than the pure valuation equilibrium, the constrained mixed equilibrium, and the unconstrained mixed equilibrium;
ii) for parameter values such that the only market equilibrium is the constrained mixed equilibrium, this equilibrium is less efficient than no valuation and trading all assets without valuation;
iii) for parameter values such that the only market equilibrium is the pure valuation equilibrium, if

$$c \ge c^{Eff} := \frac{(1 - \lambda\bar{\chi}) R - (1 - \lambda) D^b - (1 - \bar{\chi}) \lambda D^g}{\bar{\chi}R}$$

$$\tag{14}$$

then the market equilibrium is less efficient than no valuation and trading all assets without valuation.



Figure 2: The region where funding without valuation is more efficient than equilibria with valuation

#### 6.3 Application to a financial crisis

In this section, we interpret a switch from an equilibrium without valuation to an equilibrium with valuation as a financial market collapse. While the circumstances in which a collapse is possible are exogenous in the model, we discuss two scenarios in which parameters might naturally change over time so as to lead the market into the region of multiplicity or collapse. With the temporal interpretation, the results of the previous section imply that collapse is inefficient when it occurs. The next section discusses what policies would address this inefficiency.

Financial market booms and crashes often follow new real investment opportunities or the development of new financial assets (Kindleberger (2000)). Because such investment opportunities are new, there is limited evidence about which assets will pay off well and which poorly, so that valuation beyond a certain point is costly to impossible and, conditional on certain characteristics, all investment is funded or assets traded. However, two factors can change. First, over time specialists observe returns and may become better able to distinguish which assets will succeed and which fail, so that the cost of valuation declines. Alternatively (or additionally), since capital is flowing into the market without careful valuation, the quality of the pool of sellers of these assets may decline over time. In our model, either of these (exogenous) changes can bring about the possibility of a sudden financial collapse that has many of the features of the collapse of a bubble or a 'run' in the market. Presumably, following the collapse, over time the scarce resource that earns profits expands so that there is sufficient valuation capacity and trade volume recovers.

Consider first a decline across periods in the marginal costs of valuation, so that  $\{c_t\}$  declines over time to zero, and let the share of good assets,  $\lambda$ , be in an 'intermediate' range where the pure valuation equilibrium can exist. For high enough  $c_0$ , the market starts in a pooling equilibrium and all assets are traded. As valuation costs decline the market enters the region of multiple equilibria where the pure valuation equilibrium and the pooling equilibrium are possible. In this region, investment can collapse or boom as either equilibrium can be selected in any period.<sup>29</sup> Once valuation costs fall low enough, the market necessarily falls into the valuation equilibrium.

The movement from the pooling to the valuation equilibrium exhibits many of the stylized features of investment crashes. In particular, the increase in valuation is associated with:

- Investment collapse: The volume of transactions declines from 1 to  $\bar{\chi}\lambda$  as only sellers that can get their assets valued and who have good assets sell/are funded.
- Price collapse: Transaction prices fall from  $\frac{\lambda D^g + (1-\lambda)D^b}{R}$  to 1 (spreads or interest rates increase). This occurs because in the pooling equilibrium assets are scarce and valuation is not required to invest without losses, so sellers get high prices and marginal investment earns the opportunity cost of funds. In the valuation equilibrium, only skilled investors purchase assets, sellers compete for this limited valuation technology, and prices for assets are low as skilled investors earn profits and sellers receive their outside option.
- Nonfundamental volatility: The timing of the crash is not driven by fundamentals, but rather could be triggered by any small coordinating event, although fundamentals make collapse ultimately inevitable.
- *Credit crunch:* Some assets that would have been sold/funded in the pooling equilibrium, even some unvalued assets (and so some good assets), cannot get sold/funded in the valuation equilibrium.
- *Flight to quality:* Unsophisticated investors leave the market as the chance of buying a bad asset increases and only good assets trade.
- *Profits for sophisticated investors:* Sophisticated investors make profits/valuation capacity earns rents.

What if the share of good assets is higher so that the market ultimately switches from the pooling equilibrium to the constrained mixed equilibrium? In this case, the switch in equilibrium exhibits a (smaller) collapse in price, possibly not driven by fundamentals, a decline in trust, and profits for sophisticated investors. But trade volume does not decline, there is no credit/funding crunch, and there is no flight to quality.

The second scenario for a financial collapse is the deterioration in the average quality of assets. When a market is not doing valuation, there is no reward to a seller for higher asset quality. While the quality is exogenous in the model, in applications, the share of good assets may decline over time in the pooling equilibrium because there is no return to sellers of improving their assets in the dimension not valued.<sup>30</sup>

<sup>&</sup>lt;sup>29</sup>The date of collapse could be determined by assumptions about lack of common knowledge about fundamentals, following the literature on multiple equilibria building on Morris and Shin (1998).

<sup>&</sup>lt;sup>30</sup>As an aside, there is also no higher payment to the sellers of good assets relative to bad assets in the pure valuation equilibrium as both recieve their reservation values. This result would change if sophisticated investors sometimes valued the same asset and competed to buy it.

Consider a sequence of equilibria in which  $\{\lambda_t\}$  starts very close to one and declines and in which c is in an intermediate range where the pure valuation equilibrium is possible. In this case, if the market begins in the pooling equilibrium, as  $\lambda$  declines, first a price collapse becomes possible as the constrained mixed equilibrium becomes possible, then a collapse becomes possible as the pure valuation equilibrium becomes possible. If  $\lambda$  continues to decline, either the market can return to (or remain in) the pooling equilibrium and eventually shut down (for high c), or it enters the pure valuation equilibrium and eventually shuts down (low c).

#### 6.4 Discussion of assumptions

Before moving on, we discuss the importance of four assumptions (others are discussed elsewhere).

First, it is not essential that valuation capacity be strictly limited, but it must have increasing costs. If the cost of valuation capacity to sophisticated investors were increasing in the aggregate amount of valuation purchased, our results would be qualitatively similar (if increasing 'enough') except that the sophisticated investors would not earn rents (which instead would presumably accrue to the providers of the valuation capacity). A capacity constraint significantly simplifies the analysis.

Second, instead of assuming that valuation capacity is sunk, we could have assumed that the unsophisticated investors were small and competitive price takers as in the equilibrium concept in Dubey and Geanakoplos (2002). This would lead to the same regions, would not require that sophisticated investors be competitive, but obviously would change the importance of commitment for policy, discussed next. Without sunk valuation, competition in prices by deep-pocketed unsophisticated investors could eliminates the possibility of multiple equilibria (but not change outcomes in the region in which the valuation equilibrium is unique but funding without valuation more efficient).

Fourth, the assumption that the investor gets all the surplus when matched with a seller known to be good is important only for the equations that determine where different regions occur (as long as the investor gets some of the surplus) and for what parties earn rents in the valuation equilibria. This follows because, in equilibria with valuation, sophisticated buyers are making sellers weakly prefer to sell to them, so changing this to a strict condition does not change the qualitative results.

Finally, the fact that valuation is not observed – that a seller with a previously-valued asset is indistinguishable from a seller with a valued and rejected asset – is critical. But as noted, it is also incentive compatible for the buyer-seller pair with a known-bad asset. More generally, sophisticated investors prefer equilibria with valuation and unsophisticated investors are indifferent as they make no pure profits. Sellers prefer the equilibrium without valuation. This ordering makes it suspect that investor groups that self regulate and share information, such as through industry-wide credit bureaus, actually share this type of information.<sup>31</sup>

<sup>&</sup>lt;sup>31</sup>It seems more likely that credit bureaus, like ratings agencies, simply segment assets into markets, each of which is either in a pooling equilibrium or valuation equilibrium where buyers do or do not investigate beyond the credit check. It is notable that credit bureaus typically do not reveal the identity of those who conduct credit checks and so do not reveal the purpose of the check (something that an unsophisticated investor would find useful).

# 7 Policy

There is the potential for efficiency-improving coordination or government policy in the regions of the parameter space for which there are multiple equilibria and for which the market delivers only the valuation equilibrium and  $c > c^{Eff}$ . Given that the model omits any social benefits of private information and is solved as a rational expectation equilibrium, it is worth emphasizing that this section studies optimal policy in the model not the real world.

To begin, why does the market not deliver the more efficient equilibrium? The first answer is that valuation has social costs greater than its private benefits – it creates information on which adverse selection can occur. Thus a first approach to optimal policy is to tax valuation or eliminate adverse selection.

One optimal policy is to tax units of valuation capacity with tax,  $\tau$ , so that the use of valuation is deterred where it is inefficient, which is any  $\tau$  such that

$$\tau + c \ge \left\{ \begin{array}{cc} \lambda \left(\frac{D^g}{R} - 1\right) & \text{if} \quad c \ge c^{Eff} \text{ and } \lambda \le \frac{R - D^b}{\left(D^g - D^b\right) - \bar{\chi}(D^g - R)} \\ \frac{1}{1 - \lambda \bar{\chi}} \lambda \left(1 - \lambda\right) \frac{D^g - D^b}{R} & \text{if} \quad c \ge c^{Eff} \text{ and } \lambda \ge \frac{R - D^b}{\left(D^g - D^b\right) - \bar{\chi}(D^g - R)} \end{array} \right\}$$

This tax ensures the pooling equilibrium wherever it is ex ante socially efficient. This of course has the real-world problems of both distinguishing this type of valuation from other types of valuation (such as about value that is common across assets) and monitoring and observing valuation.

Eliminating the adverse selection that follows from valuation is more effective in that it allows the use of valuation when privately efficient while eliminating its social loss. Such a policy is at odds with the assumptions of the model and not straightforward to implement given agent's incentives. But consider making it public knowledge that an asset had been valued. Then valued and rejected assets would remain unsold. Unvalued assets would be sold to unsophisticated investors at the pooling equilibrium price (iff  $\geq 1$ ). And unvalued sellers would only approach sophisticated investors if the expected value of their outside options after valuation were at least equal to the price available without valuation, which cannot happen unless either sophisticated investors pay fees to unvalued sellers prior to valuation or sophisticated sellers can commit to posted prices.<sup>32</sup> Considering the case of commitment to posted prices, the sophisticated investor would post the minimum price to attract unvalued assets, which is  $P^g$  such that

$$\lambda P^g + (1-\lambda) \ge \frac{\lambda D^g + (1-\lambda) D^b}{R}.$$

Valuation would be undertaken when  $J^S \ge 0$  (with  $P^U = 1$ ), which is when

$$c \le \lambda \left(\frac{D^g}{R} - 1\right). \tag{15}$$

$$\lambda \left(\frac{\lambda D^g + (1-\lambda) D^b}{R}\right) + (1-\lambda) + fee = \frac{\lambda D^g + (1-\lambda) D^b}{R}$$

 $<sup>^{32}</sup>$ The ability of sophisticated investors to commit to prices would not change the results of sections 5 and 6. A fee would have the same implications as the price described in the main text and would satisfy:



Figure 3: The region where eliminating adverse selection delivers the efficient outcome

This boundary lies above the boundary for efficiency of the pooling equilibrium with anonymity (equation (14); equations (14) and (15) converge as  $\bar{\chi} \to 1$ ).

In Figure 3, the solid lines delineate the three equilibria when there is price commitment (dotted lines delineate the baseline model regions; and the dash lines delineates the boundary of the region in which funding without valuation is efficient). There is a larger region in which the pooling equilibrium exists, determined by equation (9) with the first inequality replaced by the complement of equation (15). There is a new type of constrained mixed equilibrium in which the price is the same as the pooling equilibrium, trade is  $\lambda \bar{\chi} + (1 - \bar{\chi})$ , and  $(1 - \lambda) \bar{\chi}$  bad assets are not traded, determined by equations (9) and (15). Third, there is a region of pure valuation in which the price is 1 and trade is  $\lambda \bar{\chi}$ , determined by equation (10) with the second inequality replaced by  $\lambda \leq \frac{R-D^b}{D^g-D^b}$ . Finally, there a region of no trade which covers the same region as before.

Three results follow. First, with observed valuation and without anonymity, the market equilibrium is always efficient; there are no externalities and no regions of multiplicity.<sup>33</sup> Second, some valuation is efficient for a larger set of parameters than in the original model with anonymity because valued assets that are found to be bad are not traded, and do not reduce the average quality of assets remaining after valuation. Finally, there are strict efficiency benefits to making valuation observable and eliminating anonymity if and only if original market equilibrium has valuation and unvalued assets are worth selling/funding (in Figure 3, any region with valuation in the baseline model (dotted lines) and to the right of the vertical line  $\lambda = \frac{R-D^b}{D^g-D^b}$ ).

Whether such a policy is optimal of course depends on its cost to implement. Further, such a policy is not incentive compatible given only lack of anonymity. Sophisticated investors and

<sup>&</sup>lt;sup>33</sup>This follows from similar arguments to section 6.2.

assets found to be bad have a joint incentive to hide the fact that a valuation was done.<sup>34</sup>

The second answer to why the market does not deliver the more efficient equilibrium is that unsophisticated agents do not have the ability to commit to purchase at high prices (they do not compete in contracts with commitment). If unsophisticated investors had commitment, a large unsophisticated investor could post a price equal to the price in the pooling equilibrium,  $P = \frac{\lambda D^g + (1-\lambda)D^b}{R}$ , which would ensure that the market is in the pooling equilibrium wherever it exists as a market equilibrium. This result follows from the efficiency of the pooling equilibrium.<sup>35</sup> Thus, if private agents were unable and the government were able, the government could commit to purchase at the pooling equilibrium price. Or alternately, the government could commit to insure all mortgages at the ex ante fair price for the pooling equilibrium.<sup>36</sup>

However, the ability of a large unsophisticated investor to commit does not ensure that valuation is not used outside the region of multiplicity where the efficient outcomes still involves no valuation. To ensure this, the government further has to subsidize purchases by the large investor, as for example by a proportional subsidy  $\sigma$  that implies  $P^U = (1 + \sigma) \frac{\lambda D^g + (1 - \lambda)D^b}{R}$  and  $J^S \leq 0$ . That is, if  $c \geq c^{Eff}$  and  $c \leq \lambda (1 - \lambda) \frac{D^g - D^b}{R}$ , then

$$\sigma = \frac{(1-\lambda)\left(D^g - D^b\right) - cR/\lambda}{\lambda D^g + (1-\lambda)D^b}$$

(along with ex ante commitment by a large unsophisticated agent) ensures that  $J^S < 0$  and the economy is in a pooling equilibrium wherever it is more efficient (ignoring the cost of the subsidy).

The model provides an interpretation of the government-sponsored enterprises Fannie Mae and Freddie Mac. In the market for conforming mortgages a small fraction of assets (mortgages) are 'bad' so that  $\lambda$  is close to one. For  $\lambda$  close to one, equilibria with valuation when they exist are inefficient for a wider range of parameters than when  $\lambda$  is lower. And a large investor with commitment and a subsidy that funds a large fraction at high prices can keep other from investing in valuation capacity and so optimally ensure the efficient equilibrium. One might then interpret the demise of these institutions as due to their committing to purchase at the expected present discounted value of a random (unvalued) mortgage with  $\lambda$  too low to support this as a pooling equilibrium price. In this case, the commitment to purchase (or insure) mortgages assuming no adverse selection when the market actually has valuation and adverse selection is extremely costly to the government (or GSE). It is also worth noting that, as with some other mechanisms to eliminate adverse selection, there are incentives to undermine this policy: in the pooling equilibrium, unsophisticated investors earns no rents, while in the valuation equilibrium, sophisticated investors make profits.

We conclude by considering two policies in which the government changes parameters of the market.

 $<sup>^{34}</sup>$ Sorkin (2009) describes several episodes during the US financial crisis of commerical banks valuing an investment bank for purchase in which both parties, but especially the investment bank, tried to keep secret the fact that a valuation was occurring.

<sup>&</sup>lt;sup>35</sup>Since  $P^S \ge P$  to compete, it is straightforward to verify that  $J^S < 0$  wherever the pooling equilibrium is possible, so that no sophisticated investors would invest in valuation and the multiplicity is eliminated.

<sup>&</sup>lt;sup>36</sup>It is not sufficient to insure the mortgages at an ex post fair price, since then sophisticated investors can do valuation, insure only the bad assets, and destroy the insurance scheme.



Figure 4: The effect of a higher real interest rate

First, cutting the interest rate is counterproductive. The set of (other) parameters for which equilibria with valuation are possible with a lower interest rate covers that with a higher interest rate. In contrast, raising the interest rate can held reduce valuation. These effects work by changing the present value of the information gathered by valuation, which is proportional to  $\frac{D^g - D^b}{R}$  without changing its cost. Figure 4 shows how raising the interest rate (from R = 1.10, dotted lines to R = 1.11, solid lines) reduces the size of region of multiplicity and the size of the region in which valuation can occur in conjunction with pooling. Note that the policy also increases the region in which no investment occurs.

Second, policies which reduce the difference in payoffs across assets of different quality reduce the size of the regions in which equilibria with valuation are possible. Such policies reduce the incentive to do valuation by reducing the benefits to separating the good from the bad. Figure 5 depicts how subsidizing the payout of the bad asset bad assets (from  $D^b = 1.090$  (solid line) to  $D^b = 1.095$  (dotted line)) increases the size of the pure pooling equilibrium and decreases the size of the pure valuation region, and raises the size of the region where the pooling equilibrium coexists with the constrained mixed equilibrium. This policy has some of the flavor of the TARP programs that provided funding and took some of the downside risk of private investors' asset purchases.

More generally, for any given parameterization, efficiency can be ensured through a balancedbudget subsidy ( $\sigma$ ) to ultimately bad assets that is paid for by a tax ( $\tau$ ) on ultimately good assets that satisfies

$$c \ge \lambda \left(\frac{(1-\tau)D^g}{R} - 1\right) \quad \text{if} \quad \lambda \le \frac{R - D^b}{\left(D^g - D^b\right) - \bar{\chi}(D^g - R)}$$
$$c \ge \frac{1}{1 - \lambda \bar{\chi}} \lambda \left(1 - \lambda\right) \frac{(1-\tau)D^g - (1+\sigma)D^b}{R} \quad \text{if} \quad \lambda \ge \frac{R - D^b}{\left(D^g - D^b\right) - \bar{\chi}(D^g - R)}$$



Figure 5: The effect of a higher value of bad project

where  $\tau \lambda D^g = \sigma (1 - \lambda) D^b$  ensures revenue neutrality. Of course this policy is effective because private agents are assumed to be unable to commit to a similar insurance contract. And, similar to a tax on valuation, in practice this solution blunts any incentives to buy good assets in other (unmodelled) dimensions in which valuation may be optimal.<sup>37</sup>

# 8 Conclusion

In this paper we have considered a model in which valuation creates private information about a single asset and in which the benefits of the rents of this information are captured by the informed buyer. This result stands in contrast to most research on financial markets which studies the discovery of information that is common to a class of assets and is transmitted by actions through prices. In our model, too much information is created because it creates asymmetric information and causes problems of adverse selection, while in the canonical model information tends to be under-produced and markets learn too late.

<sup>&</sup>lt;sup>37</sup>Although, if the tax and subsidy plan were explicitly balanced budget and orthogonal to the mean payoff, then such a policy could avoid diminishing the incentive to collect information about asset-class wide payoffs.

# References

- AKERLOF, G. A. (1970): "The Market for "Lemons": Quality Uncertainty and the Market Mechanism," <u>The Quarterly Journal of Economics</u>, 84(3), 488–500.
- ANGELETOS, G.-M., AND A. PAVAN (2007): "Efficient Use of Information and Social Value of Information," Econometrica, 75(4), 1103–42.
- BERK, J., AND H. UHLIG (1991): "The Timing of Information in a General Equilibrium Framework," Journal of Economic Theory, 59(1), 275–87.
- BERNANKE, B., AND M. GERTLER (1989): "Agency Costs, Net Worth, and Business Fluctuations," The American Economic Review, 79(1), 14–31.
- BOLTON, P., T. SANTOS, AND J. A. SCHEINKMAN (2011): "Cream Skimming in Financial Markets," NBER Working Papers 16804, National Bureau of Economic Research, Inc.
- BREON-DRISH, B. (2011): "Asymmetric Information in Financial Markets: Anything Goes," Stanford University Manuscript.
- BROECKER, T. (1990): "Credit-Worthiness Tests and Interbank Competition," <u>Econometrica</u>, 58(2), 429–452.
- CABALLERO, R. J., AND A. KRISHNAMURTHY (2003): "Collateral Constraints and the Amplification Mechanism," Journal of Economic Theory, 111(2), 277–92.
- (2008): "Collective Risk Management in a Flight to Quality Episode," <u>Journal of</u> Finance, 63(5), 2195–2236.
- CHANG, B. (2011): "Adverse Selection and Liquidity Distortion in Decentralized Markets," Manuscript Northwestern University.
- DANG, T. V. (2008): "Bargaining with Endogenous Information," <u>Journal of Economic</u> Theory, 140(1), 339–54.
- DANG, T. V., G. GORTON, AND B. HOLMSTROM (2009): "Opacity and the Optimality of Debt for Liquidity Provision," Manuscript Yale University.
- DELL'ARICCIA, G., AND R. MARQUEZ (2006): "Lending Booms and Lending Standards," Journal of Finance, 61(5), 2511–2546.
- DEMARZO, P. M. (2005): "The Pooling and Tranching of Securities: A Model of Informed Intermediation," The Review of Financial Studies, 18(1), 1–35.
- DEMARZO, P. M., AND D. DUFFIE (1999): "A Liquidity-Based Model of Security Design," Econometrica, 67(1), 65–99.
- DIAMOND, D. W., AND P. H. DYBVIG (1983): "Bank Runs, Deposit Insurance, and Liquidity," The Journal of Political Economy, 91(3), 401–419.
- DUBEY, P., AND J. GEANAKOPLOS (2002): "Competitive Pooling: Rothschild-Stiglitz Reconsidered," The Quarterly Journal of Economics, 117(4), 1529–70.
- EISFELDT, A. L. (2004): "Endogenous Liquidity in Asset Markets," <u>The Journal of Finance</u>, 59(1), 1–30.
- FISHMAN, M. J., AND J. A. PARKER (2012): "Uninformed Lending and Government Support of Loan Markets," Notes, Kellogg School of Management.
- GEANAKOPLOS, J., AND A. FOSTEL (2009): "Leverage Cycles and the Anxious Economy," American Economic Review, 98(4), 1211–44.
- GORTON, G. (2010): <u>Slapped by the Invisible Hand: The Panic of 2007</u>. Oxford University Press, New York NY.

- GORTON, G., AND P. HE (2007): "Bank Credit Cycles," manuscript, Yale University.
- GORTON, G., AND G. ORDONEZ (2012): "Collateral Crises," manuscript, Yale University.
- GORTON, G., AND G. PENNACCHI (1990): "Financial Intermediaries and Liquidity Creation," The Journal of Finance, 45(1), 49–71.
- GREEN, E. J., AND R. H. PORTER (1984): "Noncooperative Collusion under Imperfect Price Information," Econometrica, 52(1), 87–100.
- GUERRIERI, V., AND R. SHIMER (2012): "Dynamic Adverse Selection: A Theory of Illiquidity, Fire Sales, and Flight to Quality," Manuscript University of Chicago.
- HELLWIG, C., AND L. VELDKAMP (2009): "Knowing What Others Know: Coordination Motives in Information Acquisition," Review of Economic Studies, 76(1), 223–51.
- HIRSHLEIFER, J. (1971): "The Private and Social Value of Information and the Reward to Inventive Activity," The American Economic Review, 61(4), pp. 561–574.
- HOLMSTROM, B., AND J. TIROLE (1998): "Private and Public Supply of Liquidity," <u>The</u> Journal of Political Economy, 106(1), 1–40.
- KINDLEBERGER, C. P. (2000): <u>Manias</u>, Panics, and Crashes: A History of Financial Crises. Wiley and Sons.
- KIYOTAKI, N., AND J. MOORE (1997): "Credit Cycles," <u>The Journal of Political Economy</u>, 105(2), 211–248.
- KURLAT, P. (2010): "Lemons, Market Shutdowns and Learning," MIT manuscript.
- MACKOWIAK, B., AND M. WEIDERHOLDT (2009): "Optimal Sticky Prices under Rational Inattention," American Economic Review, 99(3), 769–803.
- MANKIW, N. G. (1986): "The Allocation of Credit and Financial Collapse," <u>The Quarterly</u> Journal of Economics, 101(3), 455–470.
- MARIN, J. M., AND R. RAHI (2000): "Information Revelation and Market Incompleteness," The Review of Economic Studies, 67(3), 563–79.
- MORRIS, S., AND H. S. SHIN (1998): "Unique Equilibrium in a Model of Self-Fulfilling Currency Attacks," American Economic Review, 88(3), 587–97.
- (2012): "Contagious Adverse Selection," <u>American Economic Journal:</u> Macroeconomics, 4, 1–21.
- MYATT, D. P., AND C. WALLACE (2009): "Endogenous Information Acquisition in Coordination Games," Economics Series Working Papers 445, University of Oxford, Department of Economics.
- PAGANO, M., AND P. VOLPIN (2008): "Securitization, Transparency and Liquidity," CEPR Discussion Papers 7105, C.E.P.R. Discussion Papers.
- RAVIV, A. (1975): "Mathematical Models for Optimal Insurance Policy Selection," PhD Dissertation, Northwestern University.
- ROTHSCHILD, M., AND J. STIGLITZ (1976): "Equilibrium in Competitive Insurance Markets: An Essay on the Economics of Imperfect Information," <u>The Quarterly Journal of</u> Economics, 90(4), 629–649.
- RUCKES, M. (2004): "Bank Competition and Credit Standards," <u>Review of Financial Studies</u>, 17(4), 1073–1102.
- SORKIN, A. R. (2009): Too Big to Fail. Viking: New York, NY.
- STIGLITZ, J. E., AND A. WEISS (1981): "Credit Rationing in Markets with Imperfect Information," The American Economic Review, 71(3), 393–410.

- TOWNSEND, R. M. (1979): "Optimal contracts and competitive markets with costly state verification," manuscript Department of Economics, University of Chicago, 21(2).
- VELDKAMP, L. (2011): <u>Information Choice in Macroeconomics and Finance</u>. Princeton University Press, Princeton, NJ.
- ZHU, H. (forthcoming): "Finding a Good Price in Opaque Over-the-Counter Markets," Review of Financial Studies.

# Appendices

# A Proof of lemma 2

**Lemma 3** (Funding with valuation, funding capacity, and share of good assets) If  $h_i > 0$ , the sophisticated investor chooses  $f_i = P^g \lambda h_i$ , gets a share  $\lambda$  of good assets, only buys after valuation finds the asset to be good, and uses all its funding capacity.

Proof: Valuation cannot be slack since it is costly ex ante. Thus sophisticated investors choose funds at least sufficient to buy all assets found to be good when using all their valuation capacity.

Suppose that a sophisticated investor bought more assets that its capacity to do valuation. In this case, any seller going to this investor would have a positive probability of selling without valuation at a price equal to its outside option. Thus this sophisticated investor, would be approached by sellers with previously-valued assets that they know to be bad as well as by sellers that do not know the quality of their assets. Thus the share of good assets would be the same as for unsophisticated investors, the market share of good assets. Since this sector is competitive, any purchases without valuation would not make profits. At the same time the existence of such purchases reduces the share of sellers with good assets that approach the sophisticated investor, so that a unit of valuation is less likely to uncover a good asset. Thus, buying only conditional on a good valuation, which would keep known-bad assets away, is more profitable.

A sophisticated investor with funds greater than its valuation capacity can not commit to value all assets before funding and not use these additional funds. If there is valuation in equilibrium, then the share of good assets in the market for funding without valuation is less than that in the population. Thus, if only unvalued assets approached the sophisticated investor, the investor would find it profitable to fund without valuation at a price equal to the expected value of the unvalued asset sold to the unsophisticated investors (or 1 if the unsophisticated investors are not in the market P < 1). Thus,  $f_i > \lambda h_i$  cannot be an equilibrium if we are to have sophisticated investors funding only after valuation. Therefore, sophisticated investors must choose funding capacity ( $f_i$ ) equal to population share of good assets ( $\lambda$ ) times their valuation capacity ( $h_i$ ) and only sellers with unvalued assets with probability  $\lambda$  of being good approach the sophisticated seller.

### **B** The unconstrained mixed equilibrium

This equilibrium can occur if for some  $H \in (0, \bar{\chi})$ ,

$$J^{S} = 0$$

$$W^{S} = W^{U} > 0$$
(B.1)

As for the constrained mixed equilibrium, this thus requires,  $H \leq \tilde{H}$  and  $P^g = P^U = P$ . Equation (B.1) implies  $P^g = \frac{D^g}{R} - \frac{c}{\lambda} = P^U(H)$  which, together with equation (7), implies that the level of valuation capacity that gives indifference is

$$H^* = \frac{1}{\lambda} - (1 - \lambda) \frac{D^g - D^b}{R} \frac{1}{c}$$
(B.2)

Thus, this equilibrium exists when

$$H^* \in \left(0, \min\left[\bar{\chi}, \tilde{H}\right]\right].$$
 (B.3)

or

$$c > \lambda (1 - \lambda) \frac{D^g - D^b}{R}$$

$$c \leq \frac{1}{1 - \lambda \bar{\chi}} \lambda (1 - \lambda) \frac{D^g - D^b}{R}$$

$$c \leq \lambda \left(\frac{D^g}{R} - 1\right)$$

The first inequality is a strict inequality version of the first condition for the pooling equilibrium (equation (9)) and implies that valuation must be costly enough that not all sophisticated investors choose to do valuation. The second inequality is the same as the second inequality for the constrained mixed equilibrium, and so is the reverse of a 'scaled up' (by  $\frac{1}{1-\lambda\chi}$ ) version of the first inequality for the pooling equilibrium. The final inequality is a strict inequality version of the first condition for the valuation equilibrium (equation (10)). The third inequality is the same as the first inequality for the valuation equilibrium.

It is straightforward to verify there is only one  $H^*$  is unique and thus that there is at most one unconstrained mixed equilibrium for any parameter configuration.